

Michigan Technological University
Digital Commons @ Michigan Tech

**Michigan Tech Publications** 

1-1-2021

# Nitrogen fixation: A poorly understood process along the freshwater-marine continuum

Amy M. Marcarelli Michigan Technological University, ammarcar@mtu.edu

Robinson W. Fulweiler Boston University

J. Thad Scott Baylor University

Follow this and additional works at: https://digitalcommons.mtu.edu/michigantech-p

Part of the Integrative Biology Commons

#### **Recommended Citation**

Marcarelli, A., Fulweiler, R., & Scott, J. (2021). Nitrogen fixation: A poorly understood process along the freshwater-marine continuum. *Limnology And Oceanography Letters*. http://doi.org/10.1002/lol2.10220 Retrieved from: https://digitalcommons.mtu.edu/michigantech-p/15457

Follow this and additional works at: https://digitalcommons.mtu.edu/michigantech-p Part of the Integrative Biology Commons





Limnology and Oceanography Letters 2021 © 2021 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/10/21.0220

### **CURRENT EVIDENCE**

## Nitrogen fixation: A poorly understood process along the freshwater-marine continuum

#### Amy M. Marcarelli<sup>(1)</sup>,<sup>1</sup>\* Robinson W. Fulweiler<sup>(1)</sup>,<sup>2,3</sup> J. Thad Scott<sup>(1)</sup>

<sup>1</sup>Department of Biological Sciences, Michigan Technological University, Houghton, Michigan; <sup>2</sup>Department of Earth and Environment, Boston University, Boston, Massachusetts; <sup>3</sup>Department of Biology, Boston University, Boston, Massachusetts; <sup>4</sup>Department of Biology, Baylor University, Waco, Texas

#### Scientific Significance Statement

Nitrogen fixation, the conversion of di-nitrogen  $(N_2)$  gas to reactive nitrogen (N) by specialized microbes, plays an essential role in Earth's nitrogen cycle. Research in marine ecosystems has shown that the organisms who carry out this process are widely distributed. Yet, the rates and controls of  $N_2$  fixation in other aquatic ecosystems (i.e., lakes, rivers, streams, wetlands, and estuaries) have been vastly understudied and their ecological roles are poorly understood. We seek to promote new research to advance and transform understanding of the biodiversity, ecological controls, and cumulative contributions of  $N_2$  fixation across inland and coastal aquatic ecosystems. Addressing  $N_2$  fixation along the freshwater to marine continuum is necessary to understand and predict the future of global N cycling.

#### Abstract

 $N_2$  fixation is a major component of the global N cycle and has been extensively studied in open-ocean and terrestrial ecosystems. Yet rates and ecological dynamics remain virtually unknown for the inland and coastal aquatic ecosystems (lakes, wetlands, rivers, streams, and estuaries) that connect terrestrial and marine biomes. This is due to the diversity of these habitats as well as the traditional paradigm that  $N_2$  fixation rates were low to nonexistent, and therefore not important, in these ecosystems. We identify three major research themes to advance understanding of aquatic  $N_2$  fixation: (1) the biological diversity of diazotrophs and variability of  $N_2$ fixation rates, (2) the ecological stoichiometry of  $N_2$  fixation, and (3) the upscaling of  $N_2$  fixation rates from genes to ecosystems. Coordinating research across these areas will advance limnology and oceanography by fully integrating  $N_2$  fixation into ecological dynamics of aquatic ecosystems from local to global scales.

Biological  $N_2$  fixation, the microbial conversion of di-nitrogen ( $N_2$ ) gas to biologically reactive ammonium, is a fundamental biogeochemical process that facilitated the great

evolutionary events from which all modern biodiversity evolved. The availability of biologically reactive N limits productivity in ecosystems across the globe, as it likely has since

Associate editor: James B. Heffernan

<sup>\*</sup>Correspondence: ammarcar@mtu.edu

Author Contribution Statement: All authors contributed equally to the intellectual development of the ideas presented and equally drafted and edited the manuscript.

**Data Availability Statement:** Data and metadata for Fig. 1 are available at https://doi.org/10.6084/m9.figshare.14919660.v1 and for Fig. 3 at https://doi.org/10.6084/m9.figshare.14916456.v1.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

the earliest development of life on earth (Stüeken et al. 2016). Biological N<sub>2</sub> fixation evolved ca. 3.2 billion years ago (bya) (Stüeken et al. 2015), prior to the evolution of oxygenic photosynthesis 2.95 bya and the Great Oxidation Event 2.5-2.3 bya (Lyons et al. 2014), and likely helped to overcome constraints to carbon fixation and biomass production for the earliest microbial producers on earth. The origin of the nitrogenase enzyme which catalyzes N<sub>2</sub> fixation largely remains a mystery (Boyd and Peters 2013), but its structure has been highly conserved. Evidence suggests that molybdenum-nitrogenase existed 3.2 bya (Stüeken et al. 2015), with vanadium (V) and iron (Fe)-based nitrogenases emerging later (Bovd and Peters 2013). Although each form of nitrogenase has different activation energies and efficiencies, all organisms capable of N2 fixation require substantial energy to carry out the process (Hoffman et al. 2014), and they must protect nitrogenase from oxygen inhibition. Today, the nitrogenase enzymes are limited to specialized groups of prokaryotes (Zehr et al. 2003), collectively referred to as diazotrophs. Dozens of bacterial genera have the genetic coding to equip them to fix N<sub>2</sub>, and they typically can be classified as either symbionts with vascular plants or algae (Prechtl et al. 2004; Brouwer et al. 2017; Zehr and Capone 2020), heterocytous or nonheterocytous cyanobacteria (Whitton 2012), or free-living heterotrophic bacteria (Smercina et al. 2019).

Biological N<sub>2</sub> fixation is a key control point in the modern global N cycle. Currently, N<sub>2</sub> fixation represents the largest natural source of fixed N globally, and only in the past few decades has anthropogenic N<sub>2</sub> fixation via the Haber-Bosch processes grown to represent 45% of the total N fixed globally (Canfield et al. 2010; Schlesinger and Bernhardt 2013). Importantly, it appears N<sub>2</sub> fixation can occur at rates high enough to impact ecosystem N budgets at local to global scales (Fulweiler



**Fig. 1.** Twenty-eight-year publication history on  $N_2$  fixation in marine, terrestrial and freshwater environments derived from Web of Science with the topical search string "nitrogen fixation" \* "marine," "terrestrial," or "freshwater."

et al. 2007; Zehr and Capone 2020). Yet, because N<sub>2</sub> fixation influences the availability of a key limiting nutrient for biological activity (Vitousek et al. 2010; Zehr and Capone 2020), even low rates of this process can directly and/or indirectly influence ecological interactions throughout the biosphere. N<sub>2</sub> fixation controls the nutrient limiting primary producers at individual to ecosystem scales (Marcarelli and Wurtsbaugh 2006; Welter et al. 2015; Higgins et al. 2017), supports coevolved microbial consortia that couple cycling of carbon (C), N, sulfur (S), and other key elements (Bertics et al. 2010; Fulweiler et al. 2013; Dekas et al. 2018), and contributes to ecological interactions and symbioses among diazotrophs and a myriad of other organisms across all domains of life (Chapin et al. 1994; Foster and Zehr 2019; Zilius et al. 2020). We are only starting to understand the ways that diverse diazotrophs contribute to biogeochemical and ecological dynamics across global ecosystems, and the factors that control this process in space and time. Research in ocean ecosystems in particular has identified the important role of (1) the biological diversity of both autotrophic and heterotrophic microbes that fix  $N_2$  (Zehr et al. 2003; Moisander et al. 2010, 2017), (2) the stoichiometric constraints and regulation of N<sub>2</sub> fixation (Knapp 2012; Inomura et al. 2018; Wang et al. 2019), and (3) the importance of appropriate quantitative approaches for scaling N<sub>2</sub> fixation between local, regional, and global scales (Landolfi et al. 2018).

Despite the explosion of recent understanding of rates and controls of N<sub>2</sub> fixation on land and in the sea, the aquatic landscapes or aquascapes (e.g., lakes, wetlands, streams, rivers, and estuaries) in between have largely gone ignored. Oceanographers and terrestrial ecologists have studied the role of N<sub>2</sub> fixation in modern ecosystems more intensively than inland and coastal aquatic scientists (Fig. 1). The inequity of research across inland and coastal aquatic ecosystems may stem from their tremendous physical and ecological diversity compared to the open oceans and continents, which makes generalizing across these environments challenging. It may also be a product of early studies and syntheses reporting low N2 fixation rates in these ecosystems (Howarth et al. 1988)-which is likely due to methodological issues as well as high spatial and temporal variability within and across aquatic ecosystems. Essentially, a generation of aquatic scientists was told N<sub>2</sub> fixation was not important and thus it need not be measured. Research on N<sub>2</sub> fixation in lakes, streams, and wetlands has been dominated by measuring how autotrophic N<sub>2</sub> fixation rates depend on the stoichiometric imbalance of N and phosphorus (P) (Maranger et al. 2018). Observations from microcosm to whole-lake enrichment experiments (Schindler et al. 2008; Paerl et al. 2016) and along natural gradients of N and P availability and/or enrichment (Eberhard et al. 2018; Scott et al. 2019) have been used to build conceptual models of where and when N2 fixation may be important. Yet, this research lacks a fully developed theoretical framework on the underlying controls of N<sub>2</sub> fixation, and does not reflect the nutrient regeneration-uptake regime that dominates most natural ecosystems (Paerl et al. 2016). Therefore, we cannot answer

basic questions about the role of  $N_2$  fixation in these waters, such as how this process supports spatial and temporal variability in primary and secondary production, when and where autotrophic and heterotrophic diazotrophs contribute to the N cycle in inland waters, and how rates may be extrapolated to regional and global scales.

We are poised now to ask how  $N_2$  fixation varies among aquatic habitats across the aquascape due to patterns of biodiversity, nutrient delivery and transport, recycling and uptake affinity, and nutrient removal processes (Knapp 2012; Fulweiler et al. 2013; Eberhard et al. 2018). In order to reframe the paradigm of  $N_2$  fixation in aquatic ecosystems, we outline here current understanding and future opportunities across three major research themes: (1) the biological diversity of  $N_2$ -fixers and the environmental controls on  $N_2$  fixation, (2) the ecological stoichiometry of  $N_2$  fixation, and (3) scaling  $N_2$  fixation from genes to ecosystems. Addressing these critical research themes would transform our understanding of aquatic primary and secondary production, biogeochemical cycling, and provide a mechanism for upscaling rates and forecasting for ecosystem change.

### Biological diversity of diazotrophs and variability of N<sub>2</sub> fixation rates

Recent decades of research in marine ecosystems have demonstrated that both autotrophic and heterotrophic N2 fixation are more widespread than traditionally thought and are occurring under varied and sometimes unexpected environmental conditions (Benavides et al. 2018; Zehr and Capone 2020). Cyanobacteria have been the primary focus of N<sub>2</sub> fixation research across the freshwater-marine continuum and currently are thought to account for most of the N<sub>2</sub> fixation occurring in marine (Montoya et al. 2004; Shiozaki et al. 2014) and freshwater environments (Oliver et al. 2012). However, it has long been hypothesized that heterotrophic bacteria evolved the ability to fix N<sub>2</sub> prior to the evolution of oxygenic photosynthesis (Raymond et al. 2004; Schlesinger and Bernhardt 2013), and heterotrophic prokaryotes have recently been shown to account for substantial proportions of N<sub>2</sub> fixation in the ocean (Benavides et al. 2018; Dekas et al. 2018). Couple this with the increasing evidence for the importance of heterotrophic N2 fixation in both oxygendepleted and oxygen-replete water columns (Fernandez et al. 2011; Bentzon-Tilia et al. 2015) as well as aquatic sediments (Fulweiler et al. 2007; Grantz et al. 2012; Dekas et al. 2018), and it becomes clear that we have a tremendous amount to learn about the phylogenetic diversity and functional importance of heterotrophic diazotrophs.

Over the last two decades, studies across a range of aquatic habitats have employed new techniques and linked microbial community composition and activity to rate measurements to elucidate the environmental and ecological controls on  $N_2$  fixation rates and their contributions to ecosystem N budgets

(Zehr and Capone 2020). Quantification of nif gene abundance and expression has demonstrated that widely distributed unicellular cyanobacteria fix N2 at high rates (Zehr and Turner 2001; Montova et al. 2004; Moisander et al. 2010). Molecular techniques have also highlighted the potential importance of noncyanobacterial diazotrophs in the pelagic ocean and in coastal sediments (Brown and Jenkins 2014; Newell et al. 2016). Methodological issues related to gas dissolution and contamination have plagued <sup>15</sup>N measurements and the acetylene reduction assay (ARA), resulting in inaccurate, often underestimated rates of N<sub>2</sub> fixation (Mohr et al. 2010; Wilson et al. 2012). Moreover, acetylene toxicity alters heterotrophic sediment communities, including diazotrophs, suggesting that ARA may be an especially poor method for measuring N<sub>2</sub> fixation rates in benthic environments (Fulweiler et al. 2015). It is also important to note that these techniques may report net, actual, or potential rates depending on the incubation and measurement techniques, which complicates efforts to compare rates across habitats (Fig. 2a).

An additional challenge with understanding the distribution of N<sub>2</sub> fixation in the environment is that rates are typically measured over minutes to hours and in small-scale enclosures, and rates can be highly variable depending on the temporal and spatial scale of study. Spatial heterogeneity and/or "hot spots" where ideal environmental conditions for the process occur as have been observed to be caused by animal burrows (Bertics et al. 2010), cyanobacterial blooms, benthic mats and biofilms (Grimm and Petrone 1997; Eberhard et al. 2018), and animal microbiomes (Zilius et al. 2020; Fig. 2b). Temporal heterogeneity in N<sub>2</sub> fixation emerges as physiological responses of diazotrophs to environmental drivers occur in minutes to hours, while population-scale variation occurs across days to weeks (Marcarelli and Wurtsbaugh 2006; Masuda et al. 2018), and community and ecosystem variation occurs over seasons and among years (Lee and Joye 2006; Olofsson et al. 2020). N<sub>2</sub> fixation rates may show diel cycles that peak during the day or at night, depending on the energetics of the diazotroph and the adaptations they have evolved to protect nitrogenase from deactivation by oxygen (Grimm and Petrone 1997; Frischkorn et al. 2018; Masuda et al. 2018; Fig. 2c). Because of the short time scale of most measurements and the high variability in rates, comparing rates among studies and across habitats requires substantial environmental data and/or numerous assumptions. This issue is exemplified in the review by Howarth et al. (1988) where they report average annual  $N_2$ fixation rates (as g N m<sup>-2</sup> y<sup>-1</sup>) from approximately 75 studies published between 1970 and 1987 and required almost 2000 words in table footnotes to explain the assumptions for the different scaling approaches. Some of these assumptions were as simple as assuming a relevant duration for an hourly areal rate, while others required greater extension in converting volumetric to areal rates.

#### Marcarelli et al.



**Fig. 2.** High variability in reported rates is likely the result of a variety factors, none of which are mutually exclusive. Variability may arise from differences in (**a**) measurement techniques which report net (e.g., membrane-inlet mass spectrometry), actual (e.g., direct <sup>15</sup>N tracer techniques) or potential rates (e.g., ARAs); (**b**) spatial heterogeneity from "hot spots"; and (**c**) temporal variability from days to years (crab image from https://ian.umces.edu/imagelibrary/displayimage-6420.html; mangrove image from: https://ian.umces.edu/imagelibrary/displayimage-4579.html).

Advancing understanding of  $N_2$  fixation across aquatic habitats requires a comprehensive categorization of the organisms who carry out the processes, their energetic and ecological strategies, and their sensitivities to environmental constraints. Synthesis focused on the drivers and controls of aquatic  $N_2$  fixation needs to account for methodological differences, as well as spatial and temporal variability in the reported  $N_2$  fixation rates (Fig. 2). Future research should involve integrating rate measurements with metagenomics and metatranscriptomic information that links this biogeochemical process to specific microbial populations or communities. Such understanding is essential to advance the next research theme, enabling individual to population-level modeling of  $N_2$  fixation using ecological stoichiometry.

#### Ecological stoichiometry of N<sub>2</sub> fixation

Ecological stoichiometry integrates organismal nutrient requirements with the distribution, supply, and recycling of nutrients in their environment (Sterner and Elser 2002). This discipline can help elucidate nutrient limitation and control of organismal growth rates and life history patterns (Elser et al. 2003), and provide insights into the dynamic composition and function of communities through time and space (Evans-White and Halvorson 2017). The strength of ecological stoichiometry lies in a common framework for integrating energy and material constraints, which is necessary to understand and model patterns of N<sub>2</sub> fixation in the environment (Welter et al. 2015). To achieve this goal, we must first build a conceptual model that moves beyond the stoichiometric imbalance of N : P to integrate the full suite of nutrients that could control and constrain the organisms that fix N2 (Table 1). Here, we consider three different types of nutrient interactions with N<sub>2</sub> fixation: synthesis of the nitrogenase enzyme, energetic supply to carry out the N<sub>2</sub> fixation process, and supply of nutrients to build biomass using fixed N.

Nitrogenase is a two-component enzyme, with both components featuring active clusters or co-factors that are built around trace metals (Hoffman et al. 2014). The most common form of nitrogenase consists of a molybdenum (Mo)-Fe protein and an electron-transfer Fe protein. Therefore, synthesis of nitrogenase relies on supply of Mo and Fe, as well as C, N, and S, which form the base structure of the enzyme. Unsurprisingly, Mo has been identified as a limiting factor for N<sub>2</sub> fixation in the open and coastal ocean, and the supply of Fe has been identified as a key factor controlling the spatial pattern and overall contributions of N<sub>2</sub> fixation to lake, marine, and global ocean budgets (Mills et al. 2004; Glass et al. 2010). More recently, Fe has been linked to abundance of both cyanobacteria and nonphotosynthetic bacteria in streams (Larson et al. 2018). Access to Fe is determined by the environmental redox conditions that control the solubility and biological uptake rates (Ginn et al. 2017), as well as the availability of chelators including certain forms of dissolved organic matter (Nagai et al. 2006). The less common alternate forms of nitrogenase substitute either Fe or V for Mo in the MoFe protein, coupled with the same electron-transfer Fe protein (Hoffman et al. 2014), and with all three forms of nitrogenase exhibiting similar mechanisms and requirements for reactivity (Eady 1996). Therefore, it is reasonable to assume that organisms using these forms of nitrogenase would be similarly limited by environmental V and Fe supply. Nitrogenase may have high turnover rates in cells, with synthesis, inactivation, and degradation happening over periods as short as 24 h (Capone et al. 1990). Therefore, nitrogenase synthesis should also have a high P demand, due to amino acids required to carry out transcription and translation (Sterner and Elser 2002) from the suite of six or more nif genes that encode for the various components of nitrogenase (Boyd and Peters 2013).

 $N_2$  fixation is energy-intensive, requiring 16 ATP for reduction of  $N_2$  to two NH<sub>3</sub> molecules (Schlesinger and Bernhardt 2013; Hoffman et al. 2014); therefore,  $N_2$  fixation is also dependent on

Element	Nitrogenase synthesis	Energetic constraints	Biomass synthesis
Fe	Fe and MoFe protein in all forms of nitrogenase (Mo-, Fe-, V-)	Photosynthetic electron transport (photosystem I and II); energetics of Fe reducers	_
Мо	MoFe protein in Mo-type nitrogenase	_	_
V	VFe protein in V-type nitrogenase	_	_
Ρ	Transcription/translation of nitrogenase enzyme	ATP hydrolysis; necessary for substrate reduction	Structural component - nucleic acids; biomass building and growth
С	Structural component	Energetics - photosynthesis for autotrophs, organic matter supply for heterotrophs	Structural component; heterocytes have extra glycolipid and polysaccharide layer; secondary metabolites – toxin production
S	Structural component	Energetics of S chemoautotrophs	Structural component – amino acids
Ν	Structural component	Chlorophyll <i>a</i> synthesis for photoautotrophs	Structural component – amino acids, nucleic acids; Secondary metabolites – toxin production

**Table 1.** A summary of elemental roles in N<sub>2</sub> fixation.

biochemical processes and energetic nutrient transformations. Most directly, N<sub>2</sub> fixation will be limited by the supply of P for ATP hydrolysis as part of the nitrogenase mechanism. Indirectly, N<sub>2</sub> fixation rates will be dependent on whatever energetic pathways the diazotroph uses to fuel its metabolism. For autotrophs, this means that any nutrients that limit primary productivity (e.g., N, P, Fe, and other trace metals) may indirectly limit N2 fixation. Although not a nutrient, this also means that N<sub>2</sub> fixation by autotrophic diazotrophs may be inextricably linked to diel and seasonal patterns in light availability. Heterotrophic diazotroph metabolism is not just dependent on the supply of organic matter but also the quality (e.g., composition and biodegradability) of the organic material (Fulweiler et al. 2007). Furthermore, chemotrophic diazotrophs may couple nutrient cycles by using alternate substrates or terminal electron acceptors like SO<sub>4</sub> or Fe (Fulweiler et al. 2013; Dekas et al. 2018).

Finally, like all organisms, diazotrophs need varying compositions of elements to build their biomass, and those needs will vary based on their specific life-history strategies (Sterner and Elser 2002). Although diazotrophs are often assumed to exhibit stoichiometric homeostasis, particularly with regard to biomass N, they are also autotrophs which typically exhibit remarkable stoichiometric plasticity (Osburn et al. 2021). Similarly, aquatic heterotrophic microbes can be unexpectedly flexible in their P content, leading to a wide range of C : P stoichiometries across a range of taxa (Godwin and Cotner 2015). Among cyanobacteria, the various strategies and adaptations they have evolved to protect nitrogenase from oxygen deactivation and to deter grazers may also influence their biomass stoichiometry. For example, heterocyte cell walls have extra glycolipid and polysaccharide layers to reduce gas permeability (Flores and Herrero 2005), which will increase the C content and/or demand of those cells. Other cyanobacteria produce toxic secondary metabolites, which require both C and N for production (Van der Waal et al. 2009).

Building a mechanistic, stoichiometric understanding of diazotrophs can provide a path toward more accurate upscaling of rates in aquatic ecosystems. Zheng et al. (2019) recently demonstrated the power of stoichiometric imbalance for predicting  $N_2$  fixation rates using ecosystem-scale manipulations of terrestrial forests. Similar models have not been developed for the diverse autotrophic and heterotrophic diazotrophs of aquatic ecosystems, but developing them is fundamental to the next research theme of scaling  $N_2$  fixation from genes to ecosystems.

#### Upscaling N<sub>2</sub> fixation from genes to ecosystems

Decades of research on Earth's oceans and terrestrial habitats has led to the development of scalable models to derive spatially explicit global estimates of N<sub>2</sub> fixation (Xu-Ri and Prentice 2017; Wang et al. 2019). In both cases, these models are based on: (1) the environmental conditions supporting the growth of diazotrophs using trait-based approaches linked to their habitats, and (2) the stoichiometric imbalance between C, N, P, and micronutrient availability and the relative biological demand for these resources for primary production. Models have yielded spatially explicit estimates of N<sub>2</sub> fixation that range from 0–5 g N m<sup>-2</sup> yr<sup>-1</sup> in the open ocean to 0–10 g N m<sup>-2</sup> yr<sup>-1</sup> in terrestrial ecosystems (Fig. 3a). However, these efforts to synthesize global and regional N cycling generally ignore N<sub>2</sub> fixation in freshwaters and the coastal ocean because we lack the understanding of their spatial distribution and variability in rates, as well as the conceptual understanding and modeling tools necessary for accurate scaling.

The rates and spatial variability in N<sub>2</sub> fixation on the continents may be drastically underestimated by ignoring aquatic ecosystems. Global scaling of terrestrial N<sub>2</sub> fixation rates by Xu-Ri and Prentice (2017) results in relatively homogenous rates over continents, with low rates over much of the temperate, boreal, desert, and polar regions and higher rates in the tropics and sub-tropics (Fig. 3a). Yet, N<sub>2</sub> fixation rates in the aquatic ecosystems that are embedded in the terrestrial landscape, while spatially and temporally variable, can be quite high (e.g., desert stream Sycamore Creek: 8.4 mg N m<sup>-2</sup> h<sup>-1</sup>, Grimm and Petrone 1997; coastal temperate Narragansett Bay: 3.5 mg N m<sup>-2</sup> h<sup>-1</sup>, Fulweiler et al. 2007; warm temperate Lake Fayetteville: 5.1 mg N m<sup>-2</sup> h<sup>-1</sup>, Grantz et al. 2012). Inland



**Fig. 3.** (a) Spatially explicit global  $N_2$  fixation rates adapted from Xu-Ri and Prentice (2017) for terrestrial ecosystems and Wang et al. (2019) for oceans; and (b) a regional scale example of the landscape-scale heterogeneity of inland and coastal habitats in the northeastern United States (located at star in a).

water bodies cumulatively comprise substantial portions of the continental area (Fig. 3b). Globally, lakes and reservoirs > 0.002 km<sup>2</sup> in area comprise  $5 \times 10^6$  km<sup>2</sup> (ca. 3%) of the Earth's nonglaciated land area, while rivers and streams comprise  $7.7 \times 10^5$  km<sup>2</sup> (ca. 0.6%), with lakes disproportionately distributed in the northern temperate and boreal zones, and rivers most abundant in tropical and northern boreal zones (Verpoorter et al. 2014; Allen and Pavelsky 2018). Coastal estuaries comprise another  $1 \times 10^6$  km<sup>2</sup>, and the total continental shelf may be as much as 25 times that area (Rosentreter et al. 2021). Cumulatively, the contribution of diazotrophs in these habitats to global N<sub>2</sub> fluxes could be substantial even if rates are low and highly variable across those habitats, as has been shown for carbon dioxide and methane emissions (Raymond et al. 2013; Rosentreter et al. 2021).

Barriers to scaling of rates from small-scale measurements to global contributions are the extreme spatial diversity of inland and coastal aquatic habitats and resulting variability in environmental controls, as well as the diversity of organismal characteristics and ecological interactions outlined above. The aquatic habitats that select for diazotrophs are spatially diverse and dynamic, including the water column of lakes and estuaries, as well as the hard and soft benthic environments of streams, wetlands, lakes, and estuaries. For example, isolated freshwater wetlands can host free-living heterotrophic diazotrophs in their sediments that interact with diverse plant species (Rejmánková et al. 2018). These same wetlands can also be dominated by sediment-associated or floating cyanobacterial mats that include diazotrophs (Scott and Marcarelli 2012). The dominance of one condition or the other is controlled by the nutrient loading rates to the wetlands that constrain colonization by rooted aquatic macrophytes (Šantrůčková et al. 2010). Likewise, N<sub>2</sub> fixation may dynamically control other ecosystem processes. For example, N<sub>2</sub> fixation can be a major source of N to the biological communities of small streams (Marcarelli et al. 2008), and light availability controlled by terrestrial shading can control N inputs from N<sub>2</sub> fixation, leading to increasing concentrations of DON with increasing stream size (Finlay et al. 2011). Welter et al. (2015) showed that stream warming increased N<sub>2</sub> fixation which amplified the temperature dependence of primary production and ecosystem respiration by decreasing autotrophic N limitation. Thus, the ability to scale N<sub>2</sub> fixation is critical for predicting how ecosystems will respond to climate change and other anthropogenic forcing factors.

Numerous processes including atmospheric interactions, upland soil–water connections, hydrologic connectivity, water residence time, and internal nutrient cycling influence the stoichiometric imbalance of macronutrients and micronutrients available to diazotrophs across the aquascape. Moreover, the N biogeochemistry of these ecosystems is extremely complex and difficult to generalize. For example,  $\geq$  50% of lakes experience a N deficit (relative to P) in the range of 0–80 g N m<sup>-2</sup> yr<sup>-1</sup>, but  $\leq$  20% of lakes achieve a net N<sub>2</sub> fixation rate as high as 0–10 g N m<sup>-2</sup> yr<sup>-1</sup>, and N<sub>2</sub> fixation in most lakes never exceeds



**Fig. 4.** Conceptual model for scaling  $N_2$  fixation across aquascapes adapted from Seitzinger et al. (2006).

2 mg N m<sup>-2</sup> yr<sup>-1</sup> (Scott et al. 2019). Instead, most lakes appear to be net sources of N<sub>2</sub> to the atmosphere through denitrification, which has been recently observed for a large number of lakes in the upper mid-western United States (Loeks-Johnson and Cotner 2020). Although some evidence has suggested the importance of aquatic food webs in the fate of fixed N<sub>2</sub> (Patoine et al. 2006), burial or denitrification may remove more N than is added annually through N<sub>2</sub> fixation (Higgins et al. 2017) and drive ecosystems toward perpetual N deficiency. But, these phenomena remain drastically understudied and require targeted research to address both temporal and spatial variability at the ecosystem scale.

These examples highlight the role of N<sub>2</sub> fixation as an ecological process carried out by microorganisms interacting with their environment and other life. Thus, scaling N<sub>2</sub> fixation requires information on diazotroph presence and identity across habitats, and their activity driven through ecological interactions which are driven by complex physical, chemical, and biological conditions (Fig. 4). Stoichiometric models, similar to those applied for marine and terrestrial habitats, or biodiversity scaling models (Locey and Lennon 2016) may be useful for estimating diazotroph presence or absence at large spatial scales. But, discerning variability in N<sub>2</sub>-fixation activity will also require a greater focus on selection and dispersal mechanisms controlled by physical drivers such as hydrology and geomorphology. These processes determine the conditions to favor N<sub>2</sub> fixation, but may not control the fate of the fixed N. Long-term storage of fixed N or its transformation back to atmospheric N2 creates continued demand for new N2 fixation.

#### Next steps

The common view of  $N_2$  fixation as only a biogeochemical flux contributing to nutrient budgets has hamstrung our understanding of the potential for this process to influence ecosystem dynamics. Moreover, the proximate and ultimate fate of fixed N<sub>2</sub> has not been constrained for diverse habitats across the aquascape. We suggest that the net effect of N<sub>2</sub> fixation should be considered in an ecological framework as a novel input of N carried out by a specialized group of organisms that reside in ecological communities. In this framework, the imbalance between the variable biological stoichiometry of diverse diazotrophs and the stoichiometry of environmental nutrient supply control N<sub>2</sub> fixation rates, but the fates of fixed N can change ecological interactions and ecosystem trajectories. Moreover, focusing on both the rate and fate of fixed N will more accurately constrain the process and will be more constructive for scaling processes from population-level sensitives through community and ecosystem interactions and up to global fluxes. Together, we argue that this ecological framework will transform our understanding of N cycling by elucidating the fundamental role of N2 fixation in aquatic ecosystems.

#### References

- Allen, G. H., and T. M. Pavelsky. 2018. Global extent of rivers and streams. Science **361**: 585–588. doi:10.1126/science. aat0636
- Benavides, R., S. Bonnet, I. Berman-Frank, and L. Riemann. 2018. Deep into ocean  $N_2$  fixation. Front. Mar. Sci. **5**: 108. doi:10.3389/fmars.2018.00108
- Bentzon-Tilia, M., I. Severin, L. H. Hansen, and L. Riemann. 2015. Genomics and ecophysiology of heterotrophic nitrogen-fixing bacteria isolated from estuarine surface water. ASM MBio 6: e00929-15. doi:10.1128/mBio. 00929-15
- Bertics, V. J., J. A. Sohm, T. Treude, C.-E. T. Chow, D. G. Capone, J. A. Fuhrman, and W. Ziebis. 2010. Burrowing deeper into benthic nitrogen cycling: The impact of bioturbation on nitrogen fixation coupled to sulfate reduction. Mar. Ecol. Prog. Ser. 409: 1–15. doi:10.3354/meps08639
- Boyd, E. S., and J. W. Peters. 2013. New insights into the evolutionary history of biological nitrogen fixation. Front. Microbiol. 4: 201. doi:10.3389/fmicb.2013.00201
- Brouwer, P., and others. 2017. Metabolic adaptation, a specialized leaf organ structure and vascular response to diurnal N<sub>2</sub> fixation by Nostoc azollae sustain the astonishing productivity of azolla ferns without nitrogen fertilization. Front. Plant Sci. 8: 442. doi:10.3389/fpls.2017.00442
- Brown, S. M., and B. D. Jenkins. 2014. Profiling gene expression to distinguish the likely active diazotrophs from a sea of genetic potential in marine sediments. Environ. Microbiol. 16: 3128–3142. doi:10.1111/1462-2920.12403
- Canfield, D. E., A. N. Glazer, and P. G. Falkowski. 2010. The evolution of Earth's nitrogen cycle. Science **330**: 192–196. doi:10.1126/science.1186120
- Capone, D. G., J. M. O'Neil, J. Zehr, and E. J. Carpenter. 1990. Basis for diel variation in nitrogenase activity in the marine planktonic cyanobacterium, *Trichodesmium thiebautii*. Appl.

Environ. Microbiol. **56**: 3532–3536. doi:10.1128/aem.56. 11.3532-3536.1990

- Chapin, F. S., L. R. Walker, C. L. Fastie, and L. C. Sharman. 1994. Mechanisms of primary succession following deglaciation at Glacier Bay, Alaska. Ecol. Monogr. 64: 149–175. doi:10.2307/2937039
- Dekas, A. E., D. A. Fike, G. L. Chadwick, A. Green-Saxena, J. Fortney, S. A. Connon, K. S. Dawson, and V. J. Orphan. 2018. Widespread nitrogen fixation in sediments from diverse deep sea sites of elevated carbon loading. Environ. Microbiol. 20: 4281–4296. doi:10.1111/1462-2920.14342
- Eady, R. R. 1996. Structure-function relationships of alternative nitrogenases. Chem. Rev. **96**: 3013–3030. doi:10.1021/ cr950057h
- Eberhard, E. K., A. M. Marcarelli, and C. V. Baxter. 2018. Co-occurrence of in-stream nitrogen fixation and denitrification across a nitrogen gradient in a western U.S. watershed. Biogeochemistry **139**: 179–195. doi:10.1007/s10533-018-0461-y
- Elser, J. J., and others. 2003. Growth rate stoichiometry couplings in diverse biota. Ecol. Lett. **6**: 936–943. doi:10.1046/ j.1461-0248.2003.00518.x
- Evans-White, M. A., and H. Halvorson. 2017. Comparing the ecological stoichiometry in green and brown food webs: A review and meta-analysis of freshwater food webs. Front. Microbiol. 8: 1184. doi:10.3389/fmicb.2017.01184
- Fernandez, C., L. Farias, and O. Ulloa. 2011. Nitrogen fixation in denitrified marine waters. PLoS One 6: e20539. doi:10. 1371/journal.pone.0020539
- Finlay, J. C., J. M. Hood, M. P. Limm, M. E. Power, J. D. Schade, and J. R. Welter. 2011. Light-mediated thresholds in stream-water nutrient composition in a river network. Ecology **92**: 140–150. doi:10.1890/09-2243.1
- Flores, E., and A. Herrero. 2005. Nitrogen assimilation and nitrogen control in cyanobacteria. Biochem. Soc. Trans. 33: 164–167. doi: 10.1042/BST0330164
- Foster, R. A., and J. P. Zehr. 2019. Diversity, genomics, and distribution of phytoplankton-cyanobacterium single-cell symbiotic associations. Annu. Rev. Microbiol. **73**: 435–456. doi:10.1146/annurev-micro-090817-062650
- Frischkorn, K. R., S. T. Haley, and S. T. Dyhrman. 2018. Coordinated gene expression between Trichodesmium and its microbiome over day–night cycles in the North Pacific Subtropical Gyre. ISME J. 12: 997–1007. doi:10.1038/s41396-017-0041-5
- Fulweiler, R. W., S. W. Nixon, B. A. Buckley, and S. L. Granger. 2007. Reversal of net dinitrogen gas flux in coastal marine sediments. Nature 448: 180–182. doi:10.1038/nature05963
- Fulweiler, R. W., S. M. Brown, S. W. Nixon, and B. D. Jenkins. 2013. Evidence and an conceptual model for the cooccurrence of nitrogen fixation and denitrification in heterotrophic marine sediments. Mar. Ecol. Prog. Ser. **482**: 57–68. doi:10.3354/meps10240

- Fulweiler, R. W., E. M. Heiss, M. K. Rogener, S. E. Newell, G. R. LeCleir, S. M. Kortebein, and S. W. Wilhelm. 2015. Examining the impact of acetylene on N fixation and the active sediment microbial community. Front. Microbiol. 6: 418. doi:10.3389/fmicb.2015.00418
- Ginn, B., C. Meile, J. Wilmoth, Y. Tang, and A. Thompson. 2017. Rapid iron reduction rates are stimulated by highamplitude redox fluctuations in a tropical forest soil. Environ. Sci. Tech. **51**: 3250–3259. doi:10.1021/acs.est.6b05709
- Glass, J. B., F. Wolfe-Simon, J. J. Elser, and A. D. Anbar. 2010. Molybdenum-nitrogen co-limitation in freshwater and coastal heterocystous cyanobacteria. Limnol. Oceanogr. 55: 667–676. doi:10.4319/lo.2010.55.2.0667
- Godwin, C. M., and J. B. Cotner. 2015. Aquatic heterotrophic bacteria have highly flexible Fe content and biomass stoichiometry. ISME J. 9: 2324–2327. doi:10.1038/ismej. 2015.34
- Grantz, E. M., A. Kogo, and J. T. Scott. 2012. Partitioning whole-lake denitrification using in-situ dinitrogen gas accumulation and intact sediment core experiments. Limnol. Oceanogr. **57**: 925–935. doi:10.4319/lo.2012.57.4.0925
- Grimm, N. B., and K. C. Petrone. 1997. Nitrogen fixation in a desert stream ecosystem. Biogeochemistry 37: 33–61. doi: 10.1023/A:1005798410819
- Higgins, S. N., M. J. Paterson, R. E. Hecky, D. W. Schindler, J. J. Venkiteswaran, and D. L. Findlay. 2017. Biological nitrogen fixation prevents the response of a eutrophic lake to reduced loading of nitrogen: Evidence from a 46-year whole-lake experiment. Ecosystems **21**: 1088–1100. doi:10. 1007/s10021-017-0204-2
- Hoffman, B. M., D. Lukayanov, Z.-Y. Yong, D. R. Dean, and L. C. Seefeldt. 2014. Mechanism of nitrogen fixation by nitrogenase: The next stage. Chem. Rev. **114**: 4041–4062. doi:10.1021/cr400641x
- Howarth, R. W., R. Marino, J. Lane, and J. J. Cole. 1988. Nitrogen fixation in freshwater, estuarine, and marine ecosystems.
  1. Rates and importance. Limnol. Oceanogr. 33: 669–687. doi:10.4319/lo.1988.33.4part2.0669
- Inomura, K., J. Bragg, L. Riemann, and M. J. Follows. 2018. A quantitative model of nitrogen fixation in the presence of ammonium. PLoS One **13**: e0208282. doi:10.1371/journal. pone.0208282
- Knapp, A. N. 2012. The sensitivity of marine  $N_2$  fixation to dissolved inorganic nitrogen. Front. Microbiol. **3**: 374. doi: 10.3389/fmicb.2012.00374
- Landolfi, A., P. Kahler, W. Koeve, and A. Oschlies. 2018. Global marine  $N_2$  fixation estimates: From observations to models. Front. Microbiol. **9**: 2112. doi:10.3389/fmicb.2018.02112
- Larson, C. A., B. Mirza, J. L. M. Rodrigues, and S. I. Passy. 2018. Iron limitation effects on nitrogen-fixing organisms with possible implications for cyanobacterial blooms. FEMS Microbiol. Ecol. 94: fiy046. doi:10.1093/femsec/fiy046
- Lee, R. Y., and S. B. Joye. 2006. Seasonal patterns of nitrogen fixation and denitrification in oceanic mangrove habitats.

Mar. Ecol. Prog. Ser. **307**: 127–141. doi:10.3354/ meps307127

- Locey, K. J., and J. T. Lennon. 2016. Scaling laws predict global microbial diversity. Proc. Natl. Acad. Sci. 113: 5970–5975. doi:10.1073/pnas.1521291113
- Loeks-Johnson, B. M., and J. B. Cotner. 2020. Upper Midwest lakes are supersaturated with N<sub>2</sub>. Proc. Natl. Acad. Sci. **117**: 17063–17067. doi:10.1073/pnas.1921689117
- Lyons, T. W., C. T. Reinhard, and N. J. Planavsky. 2014. The rise of oxygen in Earth's early ocean and atmosphere. Nature **506**: 307–315. doi:10.1038/nature13068
- Maranger, R., S. E. Jones, and J. B. Cotner. 2018. Stoichiometry of carbon, nitrogen and phosphorus through the freshwater pipe. Limnol. Oceanogr. Lett. **3**: 89–101. doi:10. 1002/lol2.10080
- Marcarelli, A. M., and W. A. Wurtsbaugh. 2006. Temperature and nutrient supply interact to control nitrogen fixation in oligotrophic streams: An experimental examination. Limnol. Oceanogr. **51**: 2278–2289. doi:10.4319/lo.2006.51.5.2278
- Marcarelli, A. M., M. A. Baker, and W. A. Wurtsbaugh. 2008. Is in-stream  $N_2$  fixation an important N source for benthic communities and stream ecosystems? J. N. Am. Benthol. Soc. **27**: 186–211. doi:10.1899/07-027.1
- Masuda, T., G. Bernát, M. Bečková, E. Kotabová, E. Lawrenz, M. Lukeš, J. Komenda, and O. Prášil. 2018. Diel regulation of photosynthetic activity in the oceanic unicellular diazotrophic cyanobacterium Crocosphaera watsonii WH8501. Environ. Microbiol. 20: 546–560. doi:10.1111/ 1462-2920.13963
- Mills, M. M., C. Ridame, M. Davey, J. LaRoche, and R. J. Geider. 2004. Iron and phosphorus co-limit nitrogen fixation in the eastern tropical North Atlantic. Nature 429: 292–294. doi:10.1038/nature02550
- Mohr, W., T. Grosskopf, D. W. R. Wallace, and J. LaRoche. 2010. Methodological underestimation of oceanic nitrogen fixation rates. PLoS One **5**: e12583. doi:10.1371/journal. pone.0012583
- Moisander, P. H., R. A. Beinart, I. Hewson, A. E. White, K. S. Johnson, C. A. Carlson, J. P. Montoya, and J. P. Zehr. 2010. Unicellular cyanobacterial distributions broaden the oceanic  $N_2$  fixation domain. Science **327**: 1512–1514. doi:10. 1126/science.1185468
- Moisander, P. H., M. Benavides, S. Bonnet, I. Berman-Frank, A. E. White, and L. Riemann. 2017. Chasing after noncyanobacterial nitrogen fixation in marine pelagic environments. Front. Microbiol. 8: 1736. doi:10.3389/fmicb.2017. 01736
- Montoya, J. P., C. M. Holl, J. P. Zehr, A. Hansen, T. A. Villareal, and D. G. Capone. 2004. High rates of  $N_2$  fixation by unicellular diazotrophs in the oligotrophic Pacific Ocean. Nature **430**: 1027–1031. doi:10.1038/nature02824
- Nagai, T., A. Imai, K. Matsushige, and T. Fukushima. 2006. Effect of iron complexation with dissolved organic matter

on the growth of cyanobacteria in a eutrophic lake. Aquat. Microb. Ecol. **44**: 231–239. doi:10.3354/ame044231

- Newell, S. E., K. R. Pritchard, S. Q. Foster, and R. W. Fulweiler. 2016. Molecular evidence for sediment nitrogen fixation in a temperate New England estuary. PeerJ 4: e1615. doi:10. 7717/peerj.1615
- Oliver, R. L., D. P. Hamilton, J. D. Brookes, and G. G. Ganf. 2012. Physiology, blooms, and prediction of planktonic cyanobacteria. *In* B. A. Whitton [ed.], 2012. Ecology of cyanobacteria: Their diversity in space and time. Springer. doi:10. 1007/978-94-007-3855-3\_6
- Olofsson, M., I. Klawonn, and B. Karlson. 2020. Nitrogen fixation estimates for the Baltic Sea indicate high rates for the previously overlooked Bothnian Sea. Ambio **50**: 203–214. doi:10.1007/s13280-020-01331-x
- Osburn, F. S., N. D. Wagner, and J. T. Scott. 2021. Biological stoichiometry and growth dynamics of a diazotrophic cyanobacteria in nitrogen sufficient and deficient conditions. Harmful Algae **103**: 102011. doi:10.1016/j.hal.2021.102011
- Paerl, H. W., and others. 2016. It takes two to tango: When and where dual nutrient (N&P) reductions are needed to protect lakes and downstream ecosystems. Environ. Sci. Tech. 50: 10805–10813. doi:10.1021/acs.est.6b02575
- Patoine, A., M. D. Graham, and P. R. Leavitt. 2006. Spatial variation of nitrogen fixation in lakes of the northern Great Plains. Limnol. Oceanogr. **51**: 1665–1677. doi:10.4319/lo. 2006.51.4.1665
- Prechtl, J., C. Kneip, P. Lockhart, K. Wenderoth, and U. Maier. 2004. Intracellular spheroid bodies of Rhopalodia gibba have nitrogen-fixing apparatus of cyanobacterial origin. Mol. Biol. Evol. **21**: 1477–1481. doi:10.1093/molbev/ msh086
- Raymond, J., J. L. Siefert, C. R. Staples, and R. E. Blankenship. 2004. The natural history of nitrogen fixation. Mol. Biol. Evol. 21: 541–554. doi:10.1093/molbev/msh047
- Raymond, P. A., and others. 2013. Global carbon dioxide emissions from inland waters. Nature 503: 355–359. doi: 10.1038/nature12760
- Rejmánková, E., D. Sirova, S. T. Castle, J. Barta, and H. Carpenter. 2018. Heterotrophic N<sub>2</sub> fixation contributes to nitrogen economy of a common wetland sedge, *Schoenoplectus californicus*. PLoS One **13**: e0195570. doi:10. 1371/journal.pone.0195570
- Rosentreter, J. A., and others. 2021. Half of global methane emissions come from highly variable aquatic ecosystem sources. Nat. Geosci. **14**: 225–230. doi:10.1038/s41561-021-00715-2
- Šantrůčková, H., E. Rejmánková, B. Pivničková, and J. Snyder. 2010. Nutrient enrichment in tropical wetlands: Shifts from autotrophic to heterotrophic nitrogen fixation. Biogeochemistry **101**: 295–310. doi:10.1007/s10533-010-9479-5
- Schindler, D. W., and others. 2008. Eutrophication of lakes cannot be controlled by reducing nitrogen input: Results of

#### Marcarelli et al.

a 37-year whole-ecosystem experiment. Proc. Natl. Acad. Sci. **105**: 11254–11258. doi:10.1073/pnas.0805108105

- Schlesinger, W. H., and E. S. Bernhardt. 2013, *Biogeochemistry: An analysis of global change*, 3rd Edition. Academic Press.
- Scott, J. T., and A. M. Marcarelli. 2012. Cyanobacteria in freshwater benthic environments. *In* B. A. Whitton [ed.], 2012. *Ecology of cyanobacteria: Their diversity in space and time*. Springer. doi:10.1007/978-94-007-3855-3\_9
- Scott, J. T., M. J. McCarthy, and H. W. Paerl. 2019. Nitrogen transformations differentially affect nutrient-limited primary production in lakes of varying trophic state. Limnol. Oceanogr. Lett. 4: 96–104. doi:10.1002/lol2.10109
- Seitzinger, S., J. A. Harrison, J. K. Bohlke, A. F. Bouwmain, R. Lowrance, B. Peterson, C. Tobias, and G. Van Drecht. 2006. Denitrification across landscapes and waterscapes: A synthesis. Ecol. Appl. 16: 2064–2090 doi:10.1890/1051-0761 (2006)016[2064:DALAWA]2.0.CO;2
- Shiozaki, T., M. Ijichi, T. Kodama, S. Takeda, and K. Furuya. 2014. Heterotrophic bacteria as major nitrogen fixers in the euphotic zone of the Indian Ocean. Global Biogeochem. Cycles 28: 1096–1110. doi:10.1002/2014GB004886
- Smercina, D. N., S. E. Evans, M. L. Friesen, and L. K. Tiemann. 2019. To fix or not to fix: Controls on free-living nitrogen fixation in the rhizosphere. Appl. Environ. Microbiol. 85: e02546–e02518. doi:10.1128/AEM.02546-18
- Sterner, R. W., and J. J. Elser. 2002, *Ecological stoichiometry: The biology of elements from molecules to the biosphere*. Princeton Univ. Press.
- Stüeken, E. E., R. Buick, B. M. Guy, and M. C. Koehler. 2015. Isotopic evidence for biological nitrogen fixation by molybdenum-nitrogenase from 3.2 Gyr. Nature 520: 666– 669. doi:10.1038/nature14180
- Stüeken, E. E., M. A. Kipp, M. C. Koehler, and R. Buick. 2016. The evolution of Earth's biogeochemical nitrogen cycle. Earth Sci. Rev. 160: 220–239. doi:10.1016/j.earscirev.2016. 07.007
- Van der Waal, D. B., J. M. H. Verspagen, M. Lurling, E. Van Donk, P. M. Visser, and J. Huisman. 2009. The ecological stoichiometry of toxins produced by harmful cyanobacteria: An experimental test of the carbon-nutrient balance hypothesis. Ecol. Lett. **12**: 1326–1335. doi:10.1111/j.1461-0248.2009. 01383.x
- Verpoorter, C., T. Kutser, D. A. Seekell, and L. J. Tranvik. 2014. A global inventory of lakes based on high-resolution satellite imagery. Geophys. Res. Lett. **41**: 6396–6402. doi: 10.1002/2014GL060641
- Vitousek, P. M., S. Porder, B. Z. Houlton, and O. A. Chadwick. 2010. Terrestrial phosphorus limitation: Mechanisms, implications, and nitrogen-phosphorus interactions. Ecol. Appl. **20**: 5–15. doi:10.1890/08-0127.1

- Wang, W., J. K. Moore, A. C. Martiny, and F. W. Primeau. 2019. Convergent estimates of marine nitrogen fixation. Nature 566: 205–211. doi:10.1038/s41586-019-0911-2
- Welter, J. R., J. P. Benstead, W. F. Cross, J. M. Hood, A. D. Huryn, P. W. Johnson, and T. J. Williamson. 2015. Does  $N_2$  fixation amplify the temperature dependence of ecosystem metabolism? Ecology **96**: 603–610. doi:10.1890/14-1667.1
- Whitton, B. A. [ed.]. 2012, Ecology of cyanobacteria: Their diversity in space and time. Springer. doi:10.1007/0-306-46855-7
- Wilson, S. T., D. Bottjer, M. J. Church, and D. M. Karl. 2012. Comparative assessment of nitrogen fixation methodologies, conducted in the oligotrophic North Pacific Ocean. Appl. Environ. Microbiol. **78**: 6516–6523. doi:10.1128/AEM.01146-12
- Xu-Ri, and I. C. Prentice. 2017. Modelling the demand for new nitrogen fixation by terrestrial ecosystems. Biogeosciences 14: 2003–2017. doi:10.5194/bg-14-2003-2017
- Zehr, J. P., and P. J. Turner. 2001. Nitrogen fixation: Nitrogenase genes and gene expression. Methods Microbiol. **30**: 271–286. doi:10.1016/S0580-9517(01)30049-1
- Zehr, J. P., and D. G. Capone. 2020. Changing perspectives in marine nitrogen fixation. Science **368**: eaay9514. doi:10. 1126/science.aay9514
- Zehr, J. P., B. D. Jenkins, S. M. Short, and G. F. Steward. 2003. Nitrogenase gene diversity and microbial community structure: A cross-system comparison. Environ. Microbiol. 5: 539– 554. doi:10.1046/j.1462-2920.2003.00451.x
- Zheng, M., H. Chen, D. Li, Y. Luo, and J. Mo. 2019. Substrate stoichiometry determines nitrogen fixation throughout succession in southern Chinese forests. Ecol. Lett. 23: 336– 347. doi:10.1111/ele.13437
- Zilius, M., S. Bonaglia, E. Broman, V. G. Chiozzini, A. Samuiloviene, F. J. Nascimento, U. Cardini, and M. Bartoli. 2020.  $N_2$  fixation dominates nitrogen cycling in a mangrove fiddler crab holobiont. Sci. Rep. **10**: 1–14. doi:10. 1038/s41598-020-70834-0

#### Acknowledgments

We thank two anonymous reviewers for comments that greatly improved this paper. This work was supported by the National Science Foundation (award number 2015825 to JTS, AMM, and RWF, and 1451919 to AMM). This work was also partially supported by a grant from Rhode Island Sea Grant to RWF, and a grant from the National Institute of Environmental Health Sciences of the National Institutes of Health under award number 1P01ES028942 to JTS. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

> Submitted 04 January 2021 Revised 08 July 2021 Accepted 08 October 2021